CS/ECE 354 Machine Organization and Programming

Assignment 10 Due at noon on Wednesday, November 23, 1994 Grader: Gary Zipfel

For this assignment, you are to hand generate machine code for the following MAL assembly language code fragment. Starting addresses for data and instructions have been chosen arbitrarily. The first word of data is to be assigned address 0x00041000. The first instruction is to be assigned address 0x00311000. The first word of data and the first instruction are done for you. Use the format given below and also given in ~cs354- 1/public/a10.start to receive full credit for the assignment.

MAL Assembly Language Code Fragment

.data aa: .word -1 bb: .word 1:5 cc: .word 9

.text # MALcode that does __start: mult $10,$17 # not do anything. sw $2, bb la $15, aa # It is gibberish. j __start # Just assemble it line by line.

label1: lw $4, 0($29) add $12, $25, -5 b aftercode sub $12, $8, $24 bgez $16, label1 putc $25 aftercode:

Format for the Solution

address machine code TAL instruction (hex) (hex) (binary)

00041000 ffffffff 11111111111111111111111111111111 aa: .word -1 ... 00311000 01510018 00000001010100010000000000011000 mult $10, $17

What To Turn In

Complete and turn in the assignment on-line. When finished, turn in the solution by using the handin10 program by typing: ~cs354-1/bin/handin10 filename where filename is the file containing your solution. You will recieve feedback via a grade sheet.